Author's response to reviews

Title: Exosome derived from epigallocatechin gallate treated breast cancer cells suppresses tumor growth by inhibiting tumor-associated macrophage infiltration and M2 polarization

Authors:

Ji-Young Jang Dr. (jjy93@hanmail.net)
Jong-Kuen Lee Mr. (fz2jk@naver.com)
Yoon-Kyung Jeon Prof. (junarplus@chol.com)
Chul-Woo Kim Prof. (cwkim@snu.ac.kr)

Version: 6 Date: 8 August 2013

Author's response to reviews: see over
Cover letter_8 August

Thank you for comments of reviewers and receive a lot of help. I did best for complementary experiment. I marked with red color for modification

Reviewer's report 1

minor revisions:
  1. abstract, line 5: however NO DATA ARE AVAILABLE ON the influence .....  
     : Edit the line 5 in abstract
  2. abstract, line 10: cytokines BOTH WITH HIGH (IL6 and TGFb) and low (TNFα) EXPRESSION in M2....  
     : Edit the line 10 in abstract
  3. page 4 line 6: effects on tumorigenesis IN studies.....  
     : Edit the line 6 in p4
  4. page 4 line 18: in the serum OR plasma of patients....  
     : Edit the line 18 in p4
  5. Methods, page 6 line 10 please answer the question reported in brackets  
     : Edit the line 10 in p6
  6. page 10: line 10 probably should be BY BOILING TISSUE SECTIONS results,  
     : Edit the line 10 in p10
  7. page 14, the last two lines: reformulate, the sentence is not clear.  
     : Edit the last two lines in p14

Reviewer's report 2

The requested modifications have been performed.